Author’s response to reviews

Title: Medical complexity and time to lung cancer treatment - a three-year retrospective chart review

Authors:
Trine Stokstad (trine.stokstad@ntnu.no)
Sveinung Sørhaug (sveinung.sorhaug@ntnu.no)
Tore Amundsen (tore.amundsen@ntnu.no)
Bjørn Grønberg (bjorn.h.gronberg@gmail.com)

Version: 2 Date: 05 Dec 2016

Author’s response to reviews:

We have carried out essential editorial revisions as requested.

I was not sure whether it is common to add STATA-files as supplementals, and also not if the do files should be included. However, the additional files that was previously submitted, are in STATA formats .dta and .do

Yours sincerely,
Trine Stokstad